Author's response to reviews

Title: The scientific impact of the Sir Charles Bell Society in head and face medicine

Authors:

Rainer Laskawi (rlaskaw@gwdg.de)
Jaqueline Diels (rlaskaw@gwdg.de)
Paul Mulkens (rlaskaw@gwdg.de)
Mihael Podvinec (rlaskaw@gwdg.de)

Version: 2 Date: 19 March 2007

Author's response to reviews:

The article is accepted, changes only in the text describing the authors